Author's response to reviews

Title: Evidence-based ethics? On evidence-based practice and the "empirical turn" from normative bioethics

Authors:

Maya J Goldenberg (golden11@msu.edu)

Version: 3  Date: 28 October 2005

Author's response to reviews:

I have reviewed the formatting changes requested. The references in the manuscript text have been corrected (as requested) and the paper has been read over for grammatical errors. The title, abstract, and authors appear to be in order and identical to what is written in the submission system. The reference list also seems to be in proper format.